Author's response to reviews

Title: Clinical picture and treatment implication in a child with Capgras Syndrome: a case report

Authors:

Luigi Mazzone (gigimazzzone@yahoo.it)
Marco Armando (marco.armando@opbg.net)
Franco De Crescenzo (decrescenzo.franco@gmail.com)
Francesco De Maria (francesco.demaria@opbg.net)
Giovanni Valeri (giovanni.valeri@opbg.net)
Stefano Vicari (stefano.vicari@opbg.net)

Version: 5 Date: 20 October 2012

Author's response to reviews:

As suggested by the editor a native English speaker has edited the manuscript